Author’s response to reviews

Title: NEOSCOPE: A randomised Phase II study of induction chemotherapy followed by either oxaliplatin/capecitabine or paclitaxel/carboplatin based chemoradiation as pre-operative regimen for resectable oesophageal adenocarcinoma

Authors:

Somnath Mukherjee (somnath.mukherjee@oncology.ox.ac.uk)
Christopher N Hurt (hurtcn@cardiff.ac.uk)
Sarah Gwynne (Sarah.Gwynne@wales.nhs.uk)
Andrew Bateman (A.R.Bateman@soton.ac.uk)
Simon Gollins (simon.gollins@wales.nhs.uk)
Ganesh Radhakrishna (g.radhakrishna@nhs.net)
Maria Hawkins (maria.hawkins@oncology.ox.ac.uk)
Jo Canham (CanhamJ@cardiff.ac.uk)
Wyn Lewis (wyn.lewis4@wales.nhs.uk)
Heike Grabsch (heike.grabsch@mumc.nl)
Ricky Sharma (ricky.sharma@oncology.ox.ac.uk)
Wendy Wade (wendy.wade@wales.nhs.uk)
Rhydian Maggs (Rhydian.Maggs@wales.nhs.uk)
Bethan Tranter (Bethan.Tranter@wales.nhs.uk)
Ashley Roberts (Ashley.Roberts/CardiffandVale.wales.nhs.uk)
David Sebag-Montefiore (d.sebag-montefiore@leeds.ac.uk)
Tim Maughan (tim.maughan@oncology.ox.ac.uk)
Gareth Griffiths (G.O.Griffiths@soton.ac.uk)
Tom Crosby (tom.crosby@wales.nhs.uk)

Version: 3
Date: 27 January 2015

Author’s response to reviews:

Added a discussion to the abstract